review of income and wealth

Review of Income and Wealth Series 67, Number 2, June 2021 DOI: 10.1111/roiw.12464

INVESTMENT IN KNOWLEDGE-BASED CAPITAL AND PRODUCTIVITY: FIRM-LEVEL EVIDENCE FROM A SMALL OPEN ECONOMY†

BY MATTIA DI UBALDO* and IULIA SIEDSCHLAG

University of Sussex Business School Economic and Social Research Institute Trinity College Dublin

This paper examines the responsiveness of firm productivity to investment in knowledge-based capital (KBC) including a range of intangible assets such as research and development (R&D), intellectual property assets, computer software, organizational, and branding capital. A dynamic econometric model is estimated with micro-data from Ireland over the period 2006–2012. *Ceteris paribus*, the estimated average elasticity of productivity with respect to investment in KBC per employee is 0.3. In comparison to previous empirical studies, this paper goes beyond the representative firm approach and accounts for the heterogeneous behavior of firms which differ by ownership, size, export participation, and sector of activity. Further, the analysis finds that investing simultaneously in multiple KBC assets has complementary as well as substitution effects on firm productivity, with different interdependence patterns for specific investment combinations across groups of firms and sectors.

JEL Codes: F61, L22, O33

Keywords: firm heterogeneity, investment in intangible assets, productivity

1. INTRODUCTION

In recent years, there has been an increased focus on investment in knowledge-based capital (KBC) as a source of innovation and productivity

Notes: This research was part of the joint Research Programme on "Enterprise Exporting, Innovation and Productivity" undertaken by the Economic and Social Research Institute, Enterprise Ireland and the Department of Jobs, Enterprise, and Innovation. The views expressed in this paper are purely those of the authors. Results are based on analysis of strictly controlled Research Microdata Files provided by the Central Statistics Office (CSO) of Ireland. The CSO does not take any responsibility for the views expressed or the outputs generated from this research. We would like to thank Gerard Doolan, Andrew Murray, Ben Berstock, and Alan Corcoran in the CSO for valuable support with data access and clearance. We also thank an anonymous referee, Prasada Rao, the Editor, Marie Bourke, Egidio Farina, Paul Gorecki, Elizabeth Harvey, Karen Hynes, Declan Hughes, Martina Lawless, Kieran McQuinn, Garrett Murray, Niall O'Donnellan, Mary O'Mahony, Zuzanna Studnicka, and participants at research presentations at the Economic and Social Research Institute in Dublin, the Irish Economic Association Conference in Cork, the University of Michigan in Ann Arbor, the European Investment Bank in Luxembourg, and the European Economic Association Conference in Manchester for useful comments and discussions.

*Correspondence to: Mattia Di Ubaldo, University of Sussex Business School, Falmer, Brighton, UK (M.Di-Ubaldo@sussex.ac.uk)

[†]Results are based on analysis of strictly controlled Research Microdata Files provided by the Central Statistics Office (CSO). The CSO does not take any responsibility for the views expressed or the outputs generated from this research.

© 2020 The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

growth.¹ KBC comprises a broad range of intangible assets such as research and development (R&D), computer software and datasets, organizational know-how, firm-specific human capital, designs, and other intellectual property assets (Andrews and De Serres, 2012). This approach has been driven by the rapid growth of information and communication technologies (ICT) in the 1990s as a new general-purpose technology and the need to undertake complementary investments such as investment in skills and organizational change required to exploit the opportunities that ICT offered.²

Measuring investment in KBC³ and its impacts is challenging given the non-physical nature of intangible assets. To overcome this challenge, several methodological frameworks have been put forward. Among these, the mostly used is the one proposed by Corrado *et al.* (2005, 2009) known as the CHS framework. On the basis of the economic theory underpinning the optimal growth literature (Weitzman, 1976; Hulten, 1979), the authors have formalized their view that expenditures on a broad range of intangibles should be capitalized in company and National Accounts. Such expenditures have been grouped in three categories: (a) *computerized information*: knowledge codified in computer programs and databases; (b) *innovative property*: R&D and intellectual property assets such as patents, copyrights, designs, and trademarks; (c) *economic competencies*: knowledge embodied in firm-specific training, organizational know-how, and branding.

Using the CHS framework, recent research has helped to progress the quantification of investment in KBC assets at the industry level and to assess its contribution to productivity growth by exploiting growth accounting methodologies (Corrado et al., 2012, 2014, 2018; Dal Borgo et al., 2013; OECD, 2013; Niebel et al., 2017). Evidence provided by Corrado et al. (2018) shows that over the 2000–2013 period, investment in intangible capital contributed substantially to labor productivity growth in several countries including the U.S. and EU.⁴ Corrado et al. (2017) go one-step further and uncover complementarities between investing in ICT and other intangible capital assets. Further, they identify positive externalities from knowledge spillovers as a channel linking investment in KBC to productivity gains. Jona-Lasinio and Meliciani (2019) also use the CHS framework in a multi-country analysis and find that investment in intangible capital amplifies the productivity effect of participation in Global Value Chains. Finally, Chen (2018) studies the contribution of intangible capital to the international income differences. To be able to perform the analysis on a large number of countries, this work adopts a narrower definition of intangibles compared to the CHS framework, but finds that including intangible investment as a factor of production allows to explain a larger fraction of the cross-country

¹Recent international evidence is reviewed by OECD (2013) and Corrado *et al.* (2018). ²Karlsson *et al.* (2010) reviews the international evidence on the role of ICT as a new general pur-

²Karlsson *et al.* (2010) reviews the international evidence on the role of ICT as a new general purpose technology and complementary investments needed to exploit the growth opportunities ICT offer. ³Throughout this paper we use the terms knowledge-based capital (KBC) and intangible assets interchangeably.

⁴Austria, Belgium, the Czech Republic, Denmark, Finland, France, Germany, Greece, Hungary, Ireland, Italy, the Netherlands, Portugal, Slovakia, Slovenia, Spain, Sweden, the United Kingdom.

income variation, relative to analyses based on the traditional production factors only.

Firm-level evidence on the impact of investment in KBC on productivity is more limited due to constraints imposed by data availability. Most firm-level analvses have focused on the impact of R&D expenditures and more broadly innovation expenditures on innovation and productivity growth.⁵ A recent literature uses a broader definition of intangible assets and considers aggregate intangible capital as an additional input in a production function setting (Marrocu et al., 2012; Battisti et al., 2015). Chappell and Jaffe (2018) link survey and administrative firmlevel data from New Zealand and examine the association between investment in intangibles, firm characteristics, and a range of firm performance measures. They find that higher investment in intangibles is associated with higher revenue but not with the productivity. However, these contributions do not separate out the effect of the various types of intangible assets on productivity. Disaggregating intangibles by asset type is relevant given their different features and the interdependence of their effects on productivity: so far, only a small number of studies distinguish and quantify investment in specific KBC assets such as innovative property, computerized information, and economic competencies.

To the best of our knowledge, previous firm-level analyses of the relationship between productivity and investment in KBC using a production function framework have estimated static models, have mostly focused on average effects across firms, and have performed the analysis on data from large economies. Riley and Robinson (2011) exploit employer-employee data from the UK and focus on specific intangibles embedded in knowledge workers.⁶ They find a positive link between KBC assets and productivity, with organizational capital having a greater impact than R&D or IT capital. Crass and Peters (2014) examine the effects of investment in a comprehensive range of intangibles in German firms and identify R&D, branding, and firm-specific human capital as assets with strong productivity-enhancing effects. Furthermore, the authors provide evidence on the complementary effects from investment in R&D and the patent stock; investments in innovative capital and firm-specific human capital; and for investments in innovative capital and brand equity. Bontempi and Mairesse (2015), using data from Italian firms, separate intangibles depending on their accounting treatment (capitalized intangibles versus intangibles measured from expenses) and their economic nature (intellectual versus customer capital) and estimate that intangibles have a higher marginal productivity than tangible capital, with intellectual capital yielding higher returns than customer capital. Higón et al. (2017) estimate the effects of investments in R&D, advertising and human capital on total factor productivity in Spanish manufacturing firms. They find evidence of complementarities between investments in R&D and advertising, and between investment in advertising and human capital.

⁵Recent evidence is reviewed by Hall (2011), Broström and Karlsson (2017) and Roth (2019).

⁶Organizational workers (managers and marketing related occupations) as a proxy for economic competencies assets; R&D workers measuring innovative property assets; and IT workers measuring computerized information assets.

^{© 2020} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

This paper uses a panel of annual firm-level data from Ireland over the period 2006–2012 and provides novel evidence on the relationship between productivity and investment in R&D and in other knowledge-based assets such as copyrights, patents, licenses, computer software, organizational, and branding capital. The empirical analysis is underlined by a production function framework, more specifically a Cobb-Douglas production function augmented with knowledge-based capital. Building on this theoretical framework, we make three contributions to the literature. First, we model the relationship between productivity and investment in KBC in a dynamic set up to account for productivity's persistence and pathdependency. Second, in contrast with the evidence from large economies discussed above, we model the relationship between productivity and investment in KBC in the context of a small open economy, Ireland, one of the most globalized economies in the world. The small open economy modelling set up differs from the one of a large economy, which is relatively less open to trade, in that it accounts more directly for the role of international trade and inward foreign direct investment in the relationship between investment in KBC and productivity. This specific feature is relevant given that openness to trade and foreign direct investment are likely to impact on the returns to investment in KBC. Third, and related to the small open economy set up, we relax the assumption of a homogeneous behavior of a representative firm and allow the link between investment in KBC and productivity to differ across sectors (manufacturing and services) and across different groups of firms (domestic-owned and foreign-owned;⁷ small, medium-sized and large; exporters and non-exporters).

Throughout the empirical analysis, in addition to aggregate investment in KBC, we examine the marginal productivity of investments in specific KBC assets. Furthermore, we analyze the complementarity/substitutability of multiple investments in specific KBC assets.

The key results indicate that on average, over and above other factors, investment in knowledge-based capital is positively linked to firm productivity: an increase in investment in knowledge-based capital by ten per cent translates into a three per cent productivity gain. There is considerable heterogeneity across intangible assets of different nature and across firms with different characteristics. The evidence shows that productivity is most sensitive to investment in R&D intangibles assets in firms which are Irish-owned and operate in manufacturing sectors; in contrast, in foreign-owned and services firms, productivity is most strongly linked to investment in non-R&D intangible assets such as computer software and organizational and branding capital. The analysis also finds that investing simultaneously in multiple intangible assets has complementary as well as substitution effects on firm productivity, depending on specific investment combinations and firm and sector characteristics. One key finding of this analysis is that investing in R&D and IP assets enhances the productivity returns of investments in other intangible assets.

The remainder of this paper is structured as follows. Section 2 discusses the theoretical framework that underlies the empirical analysis. Section 3 describes the

⁷The analysis of the behavior and performance of foreign-owned firms relates to the activity of these firms reported in Ireland.

^{© 2020} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

data, the firm level measures of KBC used in the analysis and some key characteristics of the sample analyzed. Section 4 explains the econometric methodology and Section 5 discusses the empirical results. Finally, section 6 summarizes the main findings and discusses implications for the design of enterprise policies aiming to foster productivity growth and competitiveness.

2. Theoretical Framework

The initial point of departure for our empirical analysis is a theoretical framework based on a Cobb-Douglas production function⁸ augmented with knowledge-based capital (K) following Griliches (1979, 1980, 1998) and Pakes and Griliches (1984). We also add to the functional form human capital as in Lucas (1988), Romer (1990), and Mankiw *et al.* (1992).

In this initial analytical set up, output at the firm level is a function of observable production inputs (labor, physical capital, knowledge-based capital or intangible capital, and human capital) and an unobserved measure of efficiency assumed to be factor-neutral:

(1)
$$Q_{it} = A e^{\lambda_{it}} L_{it}^{\alpha_l} C_{it}^{\alpha_c} K_{it}^{\alpha_k} H_{it}^{\alpha_h} e^{\varepsilon_{it}}$$

 Q_{it} is output (real value added) in firm *i*, at time *t*. A is a constant and λ_{it} is the unobserved time and firm-specific rate of disembodied technical change. *L* and *C* are standard input factors, that is, labor and physical capital. K denotes knowl-edge-based (intangible capital) and H represents human capital. $\alpha_l, \alpha_c, \alpha_k, \alpha_h$ are the elasticities of output with respect to input factors.

The log-linear production function derived from equation (1) that can be easily estimated is as follows:

(2)
$$q_{it} = \alpha_0 + \lambda_{it} + \alpha_l l_{it} + \alpha_c c_{it} + \alpha_k k_{it} + \alpha_h h_{it} + \varepsilon_{it}$$

The small letters in equation (2) denote the respective log values of output and production inputs in firm *i* at time *t*. α_0 captures the average efficiency of production across all firms. ε_{it} denotes the error term that accounts for unobserved shocks and measurement errors.

Productivity measured as output per employee is obtained from Eq. (2) as follows:

(3)
$$q_{it} - l_{it} = \alpha_0 + (\alpha_l - 1)l_{it} + \alpha_c c_{it} + \alpha_k k_{it} + \alpha_h h_{it} + \lambda_{it} + \varepsilon_{it}$$

equation (3) can be further written as follows:

(4) $q_{it} - l_{it} = \alpha_0 + (\alpha_l + \alpha_c + \alpha_k + \alpha_h - 1)l_{it} + \alpha_c(c_{it} - l_{it}) + \alpha_k(k_{it} - l_{it}) + \alpha_h(h_{it} - l_{it}) + \lambda_{it} + \varepsilon_{it}$ Assuming constant returns to scale $\alpha_l + \alpha_c + \alpha_k + \alpha_h = 1$, equation (4) becomes:

(5)
$$q_{it} - l_{it} = \alpha_0 + \alpha_c (c_{it} - l_{it}) + \alpha_k (k_{it} - l_{it}) + \alpha_h (h_{it} - l_{it}) + \lambda_{it} + \varepsilon_{it}$$

⁸While alternative functional forms could be used (for example CES or translog functions), the Cobb-Douglas production function has analytical advantages as a first order-approximation to any production function. Syverson (2011) discusses in more details the choice of production functional forms.

 $q_{it} - l_{it}$: labor productivity of firm *i* at time *t*, measured as real value added per employee:

 $c_{it} - l_{it}$: physical capital per employee in firm *i* at time *t*;

 $k_{it} - l_{it}$: knowledge-based capital per employee in firm *i* at time *t*;

 $h_{it} - l_{it}$: human capital per employee in firm *i* at time *t*.

The parameter of main interest in our analysis is α_k which captures the elasticity of productivity with respect to knowledge-based capital.

We build on the production function model described above and add to equation (5) a number of features to contextualize it in line with recent advances in the literature on productivity. First, a well-established literature (see Syverson, 2011 for a review) has highlighted the role path-dependency in shaping the productivity performance of firms. Persistent differences in productivity between firms have been shown to be widespread (Bartelsman and Doms, 2000) and have been mostly attributed to firms' selection at entry (Hopenhayn, 1992; Melitz, 2003), demand channels⁹ (Moreira, 2017), learning-by-doing, innovative efforts, investment in higher quality managerial capital, as well as characteristics of the environment in which firms operate such as market structure, and competition (Syverson, 2011). To account for this productivity feature, we augment the model described by equation (5) with the lagged productivity, $(q_{it-1} - l_{it-1})$.

Second, in the static Cobb-Douglas production function model, the effect of competition on productivity is captured by an increase in the average efficiency across firms. Following on from existing evidence on the effects of competition on increasing intra-industry market shares of more efficient firms¹⁰ (see for example Syverson, 2004a; Foster et al., 2008; and Syverson, 2011 for a recent review of this evidence), we add to the model a measure of firm-specific intra-industry market share.

Third, in order to contextualize the Cobb-Douglas production model to the analysis of the relationship of investment in knowledge-based capital on productivity in a small open economy, we augment equation (5) with additional variables, which account for the engagement of firms in exporting and for the presence of foreign-owned affiliates. These variables capture both knowledge-spillovers and international competition effects.

Finally, following on from recent contributions to the literature on productivity dispersion across firms¹¹ we test whether the relationship between productivity and investment in KBC is heterogeneous across different groups of firms (domesticowned and foreign-owned; small, medium-sized and large; exporters and non-exporters); and across sectors (manufacturing and services). Firm heterogeneity implies that not all firms have the same capability to translate investment in KBC into productivity gains.

⁹Demand accumulation through consumer reputation and brand awareness. ¹⁰This mechanism is also referred to as Darwinian selection *between* firms (Syverson, 2011). However, there might additionally be within firm increases in productivity from heightened competition, due to productivity raising actions undertaken by producers. ¹¹Bartelsman and Doms (2000) review this literature. Eaton and Kortum (2002) and Melitz (2003)

have conceptualized the role of productivity dispersion on trade intensive and extensive margins. Syverson (2004b) provides evidence on productivity dispersion within industries in the U.S.

The augmented theoretical framework discussed above underlines the empirical models we estimate. Given the available data, the analysis focuses on the marginal productivity of investment in intangible assets rather than the marginal productivity of intangible capital stocks. Because the available panel data is unbalanced and it covers a short period (seven years), capital stocks of tangible, and intangible assets could be estimated with substantial measurement error. Furthermore, in our view, because the dependent variable and the other explanatory variables are flows, investment in tangible, and intangible assets is more relevant for productive efficiency. We therefore chose to estimate productivity returns to investments flows rather than capital stocks. This choice implies that, in our econometric set up, the coefficients can be interpreted as the short run returns to investing in a specific fixed asset, in terms of labor productivity (for a similar empirical approach see among others Hall et al., 2013), rather than as the user cost of capital.¹² The estimated coefficients in this case may reflect technological differences across producers (hence our focus on firm heterogeneity), and/or across fixed intangible assets, for which certain investments are more productivity enhancing than others. Alternatively, different initial levels of capital stocks across firms or intangible assets, or different depreciation rates, could also impact on the estimated coefficients.13

The next section describes the data and the measures of investment in intangibles used in the empirical analysis. In addition, we discuss patterns of investment in KBC by groups of firms and by sector, highlighting the importance of accounting for the heterogeneity of investment behavior.

3. DATA, MEASUREMENT, AND DESCRIPTIVE ANALYSIS

This analysis uses two datasets provided by Ireland's Central Statistics Office (CSO): the Census of Industrial Production (CIP) and the Annual Service Inquiry (ASI).

The CIP survey covers firms with three or more persons engaged and having their primary activity in mining and quarrying, manufacturing and utilities. The CIP contains information on location of ownership, turnover, employment and gross earnings, changes in capital assets, purchases of goods and services other than capital items. A more detailed questionnaire including information on changes in intangible assets is sent to firms with 20 and more persons engaged.

The ASI survey covers firms having their primary activity in the distribution and services sectors.¹⁴ The ASI is based on a census of firms with 20 and more

¹⁴The sectors covered by the ASI are: retail; wholesale; transportation and storage; accommodation and food; information and communication; real estate; professional, scientific, technical, administrative; and other selected services.

¹²In a production function with capital stocks, the user cost of capital interpretation implies that coefficients reflect a mixture of depreciation, capital gain/losses and net return to capital.

¹³Diminishing marginal returns to capital could imply that larger/smaller coefficients are due to different initial levels of capital stocks, across firms and/or assets. Furthermore, if a fixed asset depreciated much faster than another over a short time span, investment in the former asset could appear to be more productivity enhancing than investment in the latter.

persons and a stratified random sample for firms with less than 20 persons engaged. The data collected with a more detailed questionnaire sent to firms with 20 or more persons engaged includes the variables of interest for this analysis collected also with the CIP.

The broadest coverage of the available data on intangible assets in the CIP and the ASI is for the period 2006–2012. Taken together all data available from the CIP and ASI, our analysis is based on an unbalanced panel of annual data comprising 11,346 unique firms over the seven years period, which results in 38,647 firm-year observations.¹⁵ All monetary variables used in the analysis are deflated by the producer price indices for manufacturing industries at NACE 2-digit level and the consumer price index for the remaining sectors.¹⁶

To construct measures of investment in KBC at firm level, we follow the widely adopted methodological framework proposed by Corrado *et al.* (2005, 2009), known as the CHS framework. The CHS framework has been developed initially as a macroeconomic approach using available data for the U.S. and has subsequently been exploited to produce comparable data on investment in KBC across industries and countries. It provides a conceptual framework for measuring three types of KBC assets: (1) *computerized information*: knowledge codified in computer programs and computerized datasets, (2) *innovative property*: R&D and knowledge assets that are protected through intellectual property (IP) rights such as patents, designs, copyrights, and trademarks, and (3) *economic competencies*: knowledge embedded in a firm's human and structural resources such as firm-specific training, organizational capital, and branding.¹⁷

The CHS approach is based on expenditures on own produced intangible assets and purchased knowledge services. Following this approach, we have first extracted information on firms' own account investment in knowledge-based capital assets, available from the CIP and the ASI surveys: these are annual additions to capitalized R&D; software; copyrights, patents and licenses for intellectual property; and other fixed intangible assets. Subsequently, in order to obtain a broader measure of investment in KBC to also include purchased knowledge services, we added annual expenditures on purchased R&D services; royalties on technical know-how; and expenditure on management and marketing fees as proxy for organizational and branding capital.¹⁸ Finally, we have constructed measures of investments in KBC,

¹⁵On average each firm appears 3.4 times in the analyzed panel data. This average is due to some firms entering and exiting over the 2006–2012 period, either because of cessation of economic activity, or because of mergers and acquisitions. Other reasons could include changes in employment with downsized firms with less than 20 employees not being included in the more detailed surveys. ¹⁶Deflators for specific intangible fixed assets or price indices for gross fixed capital formation were

¹⁶Deflators for specific intangible fixed assets or price indices for gross fixed capital formation were not available to us at the time this analysis was carried out. However, capital stocks by manufacturing industry and by asset type in constant and current prices were made available recently. We therefore tested the robustness of our findings using these asset specific deflators for both tangible and intangible assets. The results are quantitatively and qualitatively similar with the estimates reported in Tables 3-5. These results are available from the authors upon request.

¹⁷Expenditures on market research and advertising.

¹⁸The expenditures data on knowledge services are available in the CIP and the ASI, in addition to the investment variables.

by aggregating the capitalized and current expenditures for own account knowledge-based assets and purchased knowledge services as follows:

- *investment in R&D*: annual capitalized R&D expenditure and expenditures for purchased R&D services;
- *investment in intellectual property*: annual capitalized expenditures on copyrights, patents and licenses and expenditures on royalties on technical know-how;
- *investment in software*: annual capitalized expenditures for computer software;
- *investment in organizational and branding capital*: expenditures on management and marketing fees;
- *investment in other intangibles*: capitalized expenditures on other intangible fixed assets;
- *total investment in intangibles*: investment in R&D, software, organizational, and branding capital, intellectual property, and other intangible assets.

Given the advantage of the granularity of our data, we analyze investment in R&D and investment in intellectual property assets (henceforth IP) as distinct investments.¹⁹

One of the key contributions of this paper is the analysis of the heterogeneous investment behavior and performance of different types of firms. Firms' decisions related to investment in KBC are likely to be influenced by their ownership status (Irish or foreign-owned), their sector of activity, their involvement in exporting and their size. This perspective is informed by previous firm-level evidence on investment in innovation²⁰ and further supported by the descriptive analysis in this section.

Table 1 explores the differences in the average amounts invested in intangibles assets and reveals systematic and statistically significant gaps across the various subgroups in which the main sample was divided. For each subsample, two sets of figures are shown: the mean value regardless of firms' investment status and, in parenthesis, the mean value computed only on the fraction of firms reporting positive investment figures. The share of firms with positive investment, also reported (Investment >0), varies substantially across subsamples and needs to be taken into account in order to compare the investment behavior accurately. Firms might, in fact, self-select into investing in intangibles, with this propensity varying between the various categories of firms analyzed.²¹

Irish-owned firms show a lower intensity²² of investment in KBC than foreign-owned firms, in all types of intangible assets: the gap is largest for IP assets, followed by investment in organizational capital and investment in R&D. The

²²Measured as investment per employee.

 $^{^{19}}Aggregating R\&D$ and IP assets does not affect the empirical findings. These results are available on request from the authors.

²⁰This evidence has been obtained with analysis based on the CDM model (Crépon et al. 1998). Recent reviews of this evidence include Mairesse and Mohnen (2010), Ruane and Siedschlag (2013), and Broström and Karlsson (2017). Siedschlag and Zhang (2015) provide evidence for Ireland.

²¹From an econometric point of view, the selection bias is not problematic since we observe both investors and non-investors in intangibles and we exploit all firms in the analysis.

	(1)	Irish	Services	Non-Exporters	SMEs	Small
	(2)	Foreign	Manufacturing	Exporter	Large	Medium and Large
Intangibles/Empl.	Mean (1) Investment >0 Mean (2) Investment >0 Diff: (1)-(2)	$\begin{array}{c} 3 \ (6,6) \\ 45.9\% \\ 84.6 \ (119) \\ 70.2\% \\ *** \ (***) \end{array}$	11.2 (26.4)42.4%35.9 (50.7) $70.5%*** (***)$	$\begin{array}{c} 6.9 \ (16.4) \\ 42.4\% \\ 39.3 \ (59.0) \\ 66.3\% \\ *** \ (***) \end{array}$	$12.7 (26.3) \\ 48.4\% \\ 80.4 (107) \\ 73.5\% \\ *** (***)$	$\begin{array}{c} 9.2 \ (20.4) \\ 45\% \\ 29.6 \ (51.0) \\ 57.9\% \\ *** \ (***) \end{array}$
R&D/Empl.	Mean (1) Investment >0 Mean (2) Investment >0 Diff: (1)-(2)	$\begin{array}{c} 0.3 \ (9.5) \\ 11.4\% \\ 5 \ (54.3) \\ 17.5\% \\ *** \ (***) \end{array}$	$\begin{array}{c} 0.5(9.3)\\ 5.8\%\\ 2.9(9.5)\\ 30.6\%\\ ***(0) \end{array}$	$\begin{array}{c} 0.3 \ (4,0) \\ 6.7\% \\ 3 \ (17,6) \\ 24.5\% \\ *** \ (***) \end{array}$	$\begin{array}{c} 1.1 \ (9.4) \\ 12\% \\ 1.4 \ (7.6) \\ 18.2\% \\ 0 \ (0) \end{array}$	$egin{array}{c} 1 \ (8.5) \ 11.4\% \ 11.4\% \ 1.4\% \ 1.4\% \ 1.4\% \ 1.4\% \ 1.4\% \ 1.6\% $
IP Assets/Empl.	Mean (1) Investment >0 Mean (2) Investment >0 Diff: (1)-(2)	$\begin{array}{c} 0.6(4.5)\\ 14.4\%\\ 55.9(216)\\ 25.6\%\\ ***(***)\end{array}$	$\begin{array}{c} 6.9 \ (60.3) \\ 11.4\% \\ 21.1 \ (70) \\ 30.1\% \\ *** \ (0) \end{array}$	$\begin{array}{c} 4 \left(32 \right) \\ 12.6\% \\ 23.8 \left(97 \right) \\ 24.6\% \\ *** \left(*** \right) \end{array}$	$\begin{array}{c} 6.9 \ (44.9) \\ 15.4\% \\ 57.1 \ (190) \\ 30\% \\ *** \ (***) \end{array}$	$\begin{array}{c} 4.9 \\ 4.6\% \\ 14.6\% \\ 18.7 \\ 10.0 \\ 19.1\% \\ *** (***) \end{array}$
Software/Empl.	Mean (1) Investment >0 Mean (2) Investment >0 Diff: (1)-(2)	$\begin{array}{c} 0.1 \ (0.8) \\ 8.9\% \\ 0.3 \ (2.2) \\ 15.1\% \\ *** \ (***) \end{array}$	$\begin{array}{c} 0.1 \ (1.1) \\ 10.6\% \\ 0.1 \ (1.5) \\ 8.2\% \\ 0 \ (0) \end{array}$	$\begin{array}{c} 0.1 & (0.9) \\ 8.7\% \\ 0.2 & (1.5) \\ 12.6\% \\ *** & (0) \end{array}$	$\begin{array}{c} 0.1 \ (1.0) \\ 9.1\% \\ 0.5 \ (2.1) \\ 21.9\% \\ *** \ (0) \end{array}$	$\begin{array}{c} 0.1 \ (0.9) \\ 8.2\% \\ 0.2 \ (1.4) \\ 12.6\% \\ *** \ (**) \end{array}$
Org. Cap/Empl.	Mean (1) Investment >0 Mean (2) Investment >0 Diff: (1)-(2)	$\begin{array}{c} 1.7 (5.4) \\ 32.2\% \\ 29.7 (54.5) \\ 54.5\% \\ *** (***) \end{array}$	$\begin{array}{c} 3.3 \ (11.6) \\ 28.2\% \\ 16.7 \ (29.4) \\ 56.7\% \\ *** \ (***) \end{array}$	$\begin{array}{c} 2.3 \ (7.7) \\ 30.2\% \\ 15.8 \ (32.5) \\ 48.6\% \\ *** \ (***) \end{array}$	4.3 (12.2) 34.9% 39.1 (73.2) 53.4% *** (***)	$2.8 (8.7) \\ 32.6\% \\ 12.4 (29.8) \\ 41.5\% \\ *** (***)$
	No. (1) No. (2)	31536 6756	27711 9863	25897 12395	35618 2674	22830 15462

equal to zero: *p < 0.10, **p < 0.05, ***p < 0.01, and 0 when null hypothesis is upheld (in parenthesis the significance of the difference in means computed for firms with positive investment values only). Source: Authors' calculations based on CIP and ASI data. Firms with 20 or more employees are included in the analysis.

@ 2020 The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

TABLE 1

intensity of investment in software is similar, but still significantly larger for foreign-owned firms. It is worth noting that despite the differences in the shares of firms investing in intangibles, the gap between Irish-owned and foreign-owned firms remains significant also when considering only investors.

On average, manufacturing firms invest more in intangibles per employee relative to service firms, although the difference is attenuated when considering only firms that report positive investment. The gap is widest for organizational capital; for IP assets and R&D the difference becomes insignificant once only the share of investors is accounted for, while for investment in software the difference is insignificant regardless of the sample considered.

Also being engaged in exporting is linked to the amount invested in intangible assets, with exporters investing more than non-exporters. The differences are all statistically significant, except for software, when comparing only firms reporting positive investment.

Across the three main size groups in which we divide all firms (small, medium, and large), the largest gap in the intensity of investment in intangibles emerges between large firms and the other two categories (grouped as SMEs), in favor of the former. Grouping medium-sized firms together with large ones and comparing them to small firms reveals differences in intangibles investment which are less sharp than if SMEs are compared to large firms.

An additional important aspect to consider when analyzing the relationship between investment in intangibles and productivity is potential reverse causality: firms invest in intangibles if they have resources to do so, such that more productive firms might be more likely to invest (Baum *et al.*, 2017). The econometric methodology we exploit in this paper allows to deal with this simultaneity issue (more on this in the next section). However, a first descriptive exploration of the association between productivity measured as real value added per employee and investment in intangibles is provided in Table 2.

Table 2 reports mean differences in value added per employee between firms that invest and firms that do not invest in intangibles, split by the various subsamples analyzed. As expected, in most cases there are statistically significant differences in productivity between investors and non-investors. While significant differences in productivity between investors and non-investors for all KBC assets exist in the case of Irish-owned firms, for all the other firm groups, investing in intangibles is related to higher productivity in at least two out of four KBC assets.

Taken together, these descriptive patterns provide motivation for our econometric analysis of the relationship between investment in KBC and productivity for all firms as well as by firm groups and sectors. The next section discusses the empirical strategy and model specifications to be estimated.

4. Econometric Methodology

4.1. Baseline Model Specification

Our empirical analysis of the relationship between investment in KBC and firm productivity is underlined by the production function theoretical model discussed in Section 2. As mentioned above, we estimate a dynamic econometric

^{© 2020} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

		MEAN DIFFE	erence in Productivi	TABLE TY Between	2 n Non-Investors (1) AND INVESTO	DRS (2)		
	(1)	(2)	(3)	(4)	(5)	(9)	(7)	(8)	(6)
	Irish Firms	Foreign firms	Manufacturing	Services	Non-Exporters	Exporters	Small firms	SMEs	Large firms
All Intangible: Mean (1) Mean (2) Diff (1)-(2)	s 139.8 161.3 ***	553.8 637.4 0	133.4 182.5 ***	192.7 326.2 ***	148.3 263.3 ***	307.9 300.3 0	180 296.3 ***	178.4 273.1 ***	311.7 332.3 0
R&D Mean (1) Mean (2) Diff (1)-(2)	152.1 130.2 **	554.5 889.7 ***	168.1 167.8 0	226.9 620.6 ***	184.1 381.1 ***	309.2 283.1 0	221 319.8 ***	212.1 313.3 ***	318.5 364.2 0
IP assets Mean (1) Mean (2) Diff (1)-(2)	154.6 119.9 ***	551.5 787.1 **	99.3 327.5 ***	244 288.9 0	198.2 188.7 0	261.3 431.5 ***	232.9 227.3 0	219.9 247.5 0	164.7 708.8 ***
Software Mean (1) Mean (2) Diff (1)-(2)	145.9 188.1 ***	623.9 548.7 0	163.3 220.3 ***	245.2 282.3 0	192.3 246.9 0	297.7 338.6 0	228 278.5 0	217.9 287.2 **	342.4 270.2 *
Org. Cap. Mean (1) Mean (2) Diff (1)-(2)	143.7 162.2 ***	587.1 633.6 0	111.3 211.2 ***	218.4 327.2 ***	171.2 256.6 ***	278.5 328.5 ***	205 288.4 ***	199.1 270.8 ***	199.4 436.5 ***
Notes: The asset, that is, mu and medium-siz $*_p < 0.10$, $*_p < 0.10$, ent The "0" ent Source: Aut	rows report mean ltiple investors α ed firms. * $p < 0.05$, *** $p <$ ries denote cases hors' calculation.	n productivity compu- ontribute to more thin < 0.01, for the null hy where fail to reject t s based on CIP and	ated for firms investin an one mean product spothesis that the me he null hypothesis th. ASI data. Firms with	ig in a specif tivity figure. an differenc at the mean 20 or more	ic intangible asset, Figures are in thou e is equal to zero. difference is equal employees are incl	regardless of Eur usands of Eur to zero.	whether a firm in o, in constant 20 nalysis.	vests also in a 10 prices. SN	mother intangible IEs includes small

@ 2020 The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

model, which accounts for the persistence of firm productivity over time and its dependence on past performance. In assessing the responsiveness of productivity to investment in KBC, we therefore account for the self-perpetuating productivity process triggered by the innate skills embodied in firms at entry, demand factors and the intangible capital already accumulated by firms in the past. In particular, we strive to eliminate the likely endogeneity arising from past productivity being correlated with current KBC investment. For these reasons, in the estimation set up we link firm productivity to its productivity performance in the previous year.²³

In addition, we control for other firm-level factors related to productivity and investment in KBC suggested by the literature on determinants of productivity. Finally, we exploit the panel nature of the data, which allows us to account for unobserved time invariant confounding factors at the firm, industry and year level.²⁴ Taken together all these factors, we estimate the following dynamic model:

(6)

$$\ln \left(VA/Empl_{ij,t} \right) = \beta_0 + \beta_1 \ln \left(VA/Empl_{ij,t-1} \right) + \beta_2 \ln \left(Intangibles/Empl_{ij,t} \right) + \beta_4 \ln \left(Tangibles/Empl_{ij,t} \right) + \beta_5 \ln \left(Wage/Empl_{ij,t} \right) + \beta_5 \ln \left(Age \right)_{ij,t} + \beta_5 (For)_{ij,t} + \beta_6 \ln \left(Mark.Sh. \right)_{ij,t} + \beta_7 (Exp)_{ij,t} + \delta_t + \sigma_i + \rho_j + \mu_{ijt}$$

The dependent variable, firm productivity²⁵, is measured as real value-added per employee²⁶ Ln ($VA/Empl_{ij,t}$) of firm *i* in NACE 2-digit sector *j* and year *t*, taken in its natural logarithm. The lagged value of the dependent variable accounts for the dynamic process driving firm productivity. The main variable of interest is

²³Controlling for past productivity over two periods instead of one leaves the results unchanged; furthermore, the second productivity lag turns out to be insignificant in the aggregate sample and most of the subsamples. These estimates are available on request.
²⁴A possible source of concern could be measurement error due to firms' mis-reporting of capital-

²⁴A possible source of concern could be measurement error due to firms' mis-reporting of capitalized expenditures. Since we have no reason to believe that this mis-reporting might not be random, this would bias the estimated coefficients towards zero. ²⁵We use labor productivity instead of alternative measures of firm efficiency, such as TFP mainly

²⁵We use labor productivity instead of alternative measures of firm efficiency, such as TFP mainly because of data limitations. While information on investment at firm-level is available, information on physical capital stock at firm level is not available. Estimates obtained with the perpetual inventory method appear to be affected by measurement error most likely due to assumptions related to initial stocks and depreciation rates and the lack of sufficiently long time series.

²⁶Given that information on physical output is not available, the productivity is measured on the basis of deflated sales. This implies that the productivity measure, particularly in the case of foreign-owned firms may be distorted by transfer pricing. Since transfer pricing is not observed in the data, to ensure that revenue distortions do not impact systematically the results of this analysis, the key regressions are run separately for Irish-owned and foreign-owned firms.

investment in intangible assets per employee,²⁷Ln (Intangibles/Empl_{ij,t}), whose association with productivity is identified by the parameter β_2 .

We augment the theoretical model discussed in Section 2 with the time-varying explanatory variables, suggested by the literature on determinants of firm productivity:²⁸ the amount invested in tangibles per employee, the wage per employee (a proxy for a firm's human capital, or skill intensity), the age of the firm, an ownership indicator, a firm's market share (a proxy for a firm's market power), and a binary variable identifying export participation as a proxy for exposure to international competition. The remaining variables, δ_i , σ_i and ρ_i , denote a set of time, firm, and NACE 2-digit industry²⁹ fixed effects.³⁰

The dynamic-panel setting applied to the available data requires instrumenting the lagged dependent variable $Ln(VA/Empl_{ii,t-1})$ to circumvent the correlation of this regressor with the residual of the model (Nickell, 1981). For this purpose, we apply a generalized method of moments (GMM) estimation, exploiting further lags of the dependent variable to instrument $Ln(VA/Empl_{ij,t-1})$. After verifying the correct specification of the model,³¹ we make a choice in favor of the system-GMM estimator (Blundell and Bond, 1998). The system GMM jointly estimates the dynamic model both in differences and in levels,³² using lagged levels as instruments for the regression in differences (from t-2 to t-6) and lagged differences as instruments for the regression in levels (dated t-1).³³ Estimating the model in both differences and levels addresses the weak instrument problem arising from using lagged-levels of persistent explanatory variables as instruments for the regression in differences, other than allowing the use of a larger portion of the sample.

²⁷It can be argued that past investment in intangibles could have an impact on productivity beyond the effect of current investment, for reasons such as learning-by-doing or persistence of innovative efforts (Piva and Vivarelli, 2007). For this reason, we attempted to extend the dynamics of our model adding the lags of intangibles to the specification but failed to estimate any significant effect from past intangibles over and above the effect of current intangibles.

See Syverson (2011) for a review of the recent literature on determinants of productivity and Arrighetti et al. (2014) for an analysis of the determinants of investment in intangibles.

The NACE dummies are estimated only for a handful of firms which change sector over the period we analyse. ³⁰Details of all variables and data sources are given in Table A1 in the Appendix.

³¹To guide the choice of the estimator and reassure about the correct specification of the GMM model, we compare results from a pooled-OLS model, a fixed-effects "within" estimator, a difference-GMM and a system-GMM estimator. While being both biased in a dynamic panel setting, the POLS and the FE estimators can be taken as, respectively, an upper and a lower bound estimate of the lagged dependent variable coefficient. Bond (2002) suggests that the GMM estimate of the lagged dependent variable should lie in between the POLS and the FE estimates, and that the system-GMM is to be preferred to the difference-GMM in case the latter estimate is close or below the FE estimate: this rule-of-thumb applies neatly in our context and suggests that the system-GMM is the appropriate estimator. These results are available on request from the authors.

However, an assumption of this approach is that changes in the instrumental variables are uncorrelated with the fixed effects: this might appear as a strong assumption in our context, but given the similarity of results between the difference-GMM (which doesn't rely on this assumption) and the system-GMM estimators, we consider this shortcoming as not very worrisome. ³³In the interest of space, we report results with one set of lags used as instruments (the *t*-2 to *t*-6

lags in levels for the differences regressions and the t-1 lag in differences for the level regressions). However, we estimated all regressions in this analysis exploiting all the possible combinations of lags (starting from t-2 and ending at t-6), in order to reassure about the robustness of the system-GMM estimates. The results are extremely similar across all lags specifications and are available on request from the authors.

The exogeneity of the lags exploited in the instrumentation is confirmed by the Arellano-Bond test for serial autocorrelation, which fails to reject the null of no first-order correlation in the residuals.³⁴ In addition, we report the Hansen J-test of the null hypothesis that the over-identifying restrictions are valid: we always fail to reject the null hypothesis, confirming the validity of the instrumentation procedure. In order to minimize the shortcomings associated with instruments over-proliferation, we present results that rely on the collapsed set of instruments obtained applying the method proposed by Roodman (2009). This method reduces drastically the number of instruments in estimation and the risk of over-fitting the model and obtaining inconsistent estimates (Newey and Smith, 2004).³⁵

The GMM instrumentation procedure can help to instrument additional regressors, which are likely to exhibit a correlation with the residual of the model, other than the lagged dependent variable. In this context, it could be argued that investment in intangibles is determined simultaneously with firm productivity, causing a reverse causality bias. Similarly, also the value of investment in tangibles, the average wage per employee and a firm's market share are likely to be endogenous to productivity due to a simultaneity issue. To circumvent these endogeneity concerns we instrument investment in intangibles, investment in tangibles, wage per employee and the market share with their past values, exploiting the same lag structure exposed for the instrumentation of the lagged dependent variable.

4.2. Testing for Complementarity/Substitution of Multiple Investments in KBC Assets

In this paper, we go beyond the analysis of independent relationships between investment in intangible assets and firm productivity and examine whether interdependence might exist between simultaneous investments in multiple intangibles in their effect on productivity. Firms are likely to combine their resources and the marginal return from investing in one intangible asset might be higher if firms invest contemporaneously in another asset. In this case, the two investments are considered to be complementary; in the opposite case (i.e. of lower marginal returns), the two investments are seen as substitutes.

Previous studies exploring complementarity and substitution across investment in intangibles reported evidence of complementarities between R&D and patents, patents and marketing, patents and human capital (Crass and Peters, 2014); as well as complementarities between advertising and R&D and advertising and human capital (Higón *et al.*, 2017).

These papers, however, analyzed samples of firms pooled together, while we allow for the relationship between investment in intangibles and productivity to be different for different groups of firms and sectors. We show that analyzing the aggregate sample only misses important relevant results on the interplay of investments

 $^{^{34}}$ This implies that lags starting at *t*-2 are uncorrelated with the residual and are valid instruments for the lagged dependent variable.

³⁵We provide more details in a Technical Appendix available from the authors. In this Appendix we also provide estimates based on the principal components instrumental variables reduction method proposed by Bontempi and Mammi (2015), as a robustness check.

in the specific intangibles assets and argue, once again, that firm heterogeneity is a key factor in the link between investment in intangibles and productivity.

A further difference with respect to previous studies is methodological. Both Crass and Peters (2014) and Higón *et al.* (2017) apply the methodology of Carree et al. (2011) in order to investigate the existence of complementarity or substitutability effects. Carree et al. (2011) estimate the econometric model with interaction terms representing all the possible cross combinations of investments. Our approach is simpler and offers results whose interpretation, while being different, is in our view more convenient. We interact each (continuous) investment variable with a binary variable identifying whether firms invest contemporaneously in another intangible asset. The interactions allow us to explore how investing in a certain asset (identified by the dummy variable) influences the relationship between investing in each of the other intangible assets and firm productivity. This approach also allows us to keep the estimation procedure lighter, and not to overflow the GMM setting with a very large number of interaction terms, which would require instrumentation with past lags.

In order to test the existence of complementarity or substitutability between investments in various intangibles, we augment the econometric model described by equation (6) with a series of interaction terms as follows:

(7)

$$\ln \left(VA / Empl_{ij,t} \right) = \beta_0 + \beta_1 \ln \left(VA / Empl_{ij,t-1} \right) + \beta_2 \ln \left(Int(A) / Emp_{ij,t} \right) + \beta_3 int(B)_{ij,t} + \beta_4 \ln \left(Int(A) / Emp_{ij,t} \right) * int(B)_{ij,t} + [\dots] + \beta_5 \ln \left(Tang / Emp_{ij,t} \right) + \beta_6 \ln \left(Wage / Emp \right)_{ij,t} + \beta_7 \ln \left(Age \right)_{ij,t} + \beta_8 (For)_{ij,t} + \beta_9 \ln (Mark.Sh.)_{ij,t} + \beta_{10} (Exp)_{ii,t} + \delta_t + \sigma_i + \rho_i + \mu_{ii,t}$$

To capture complementarity/substitution effects, the amount invested in a certain intangible (say, A) is interacted with a binary variable taking value 1 if the firm is contemporaneously investing also in another intangible (say, B). The coefficient β_4 on the interaction term captures whether the contemporaneous investment in intangible B has a positive or negative effect on the relationship between investment in intangible A and firm productivity. We interpret a positive β_4 coefficient as indicating that investment in intangible B enhances the effect of investment in intangible A (i.e. the two investments are complementary), and a negative β_4 coefficient indicating that investment in intangible B is a substitute for investment in intangible A. We include in the model described by equation (7) investments in all specific intangible assets available in our dataset (R&D, IP assets, software, organizational and branding capital, and other intangibles), together with all pair-wise interactions.

5. Empirical Results

In this section we report the estimates of models (6) and (7), exploring the relation between investment in KBC assets and firm productivity. We start by presenting the findings for the aggregate of all intangibles; next, we analyze the productivity returns linked to investments in the specific KBC assets. Finally, we

^{© 2020} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

discuss the results of our analysis of complementarity/substitutability of pair-wise combinations of investments in KBC.

5.1. Investment in All Intangible Assets and Firm Productivity

Tables 3 and 4 analyze the relationship between investment in all intangibles and firm productivity across various firm groups.

Before exploring the effect of investment in intangibles, in line with a large literature, our estimates confirm that a positive and significant determinant of firm productivity is its past performance. This result is found consistently across all the specifications estimated and firm groups analyzed. Persistence is rather high, with a ten percent higher productivity reflected, on average for all firms, in 5.4 percent higher productivity in the next time period.

The key finding of this analysis, in line with many other empirical studies (Riley and Robinson, 2011; Marrocu *et al.*, 2012; Crass and Peters, 2014; Higón *et al.*, 2017), is that investing in KBC is positively and significantly associated with firm productivity: an increase in investment in KBC per employee by ten percent translates into a 2.8 percent higher productivity.³⁶ This result, statistically robust and economically sizeable, is found when all firms are analyzed. However, as shown in Table 3, important heterogeneity exists across various groups of firms.

As mentioned in the Introduction, a novelty of our study is the separate analysis of indigenous and foreign-owned firms. Productivity is more responsive to investment in KBC in foreign-owned firms than in Irish-owned firms. This result suggests that productivity returns to investment in intangibles are larger in firms, which are more productive initially and are better at internalizing the returns from investment in intangible assets. An alternative (but not unrelated) explanation could lie in the fact that foreign firms might be endowed with more qualified personnel: Arrighetti *et al.* (2014) find that firm-level capabilities are an important source of heterogeneity in intangible investment, which in our context might explain the different productivity returns between indigenous and foreign-owned firms.

Contrasting manufacturing and service firms, we find that investment in KBC is positively linked to productivity for both types of firms, but the effect is statistically stronger for manufacturing firms relative to service firms. This result differs from Crass and Peters (2014) who find that, in Germany, productivity is (marginally) more responsive to investment in KBC in services than in manufacturing. In the context of our analysis for Ireland, this result can be explained by looking at the separate effects of investments in the various intangibles (Table 4 in the next subsection), of which, unlike for services, most of them are positive determinants of productivity in manufacturing firms.

Separating exporters and non-exporters shows that the elasticity of productivity with respect to KBC is marginally larger for non-exporters than for exporters. This is somewhat surprising, given that the sample of non-exporters is

³⁶The magnitude of the effects we estimate is larger than in some of the above-mentioned studies, although the figures are not directly comparable because of the different measures of KBC and empirical methodology.

^{© 2020} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

predominantly composed of indigenous firms. However, the statistical significance of these coefficients is too low to draw strong implications from this result.

Finally, across the three size classes, investment in KBC appears positively associated with productivity in small and medium-sized firms (with up to 250 employees), with the coefficient being positive but insignificant for large firms.

With respect to other covariates, many of the estimated coefficients are statistically insignificant. This is to be expected, given the presence of firm fixed effects, for which what identifies the coefficients are changes in the respective variables over time, within firm. It appears that investment in tangible capital per employee does not have a significant effect on firms' productivity, over and above the impact of investment in KBC. Unlike services, manufacturing firms with an increasing market share are more productive, possibly reflecting a different role of scale economies in manufacturing firms relative to services firms.³⁷ In both sectors, firms operating in international markets (foreign-owned and exporters) are more productive than firms serving only the domestic market. This result is in line with the international evidence.38

5.2. Investment in Specific KBC Assets and Firm Productivity

This section examines how investment in specific KBC assets relates to firm productivity and provides insights on one of the key contributions of this paper: we show that investments in the various intangibles have different effects across firm groups, proving that firm heterogeneity is a key feature for the analysis of productivity returns to investment in intangibles.

Besides investment in R&D, whose effects have been investigated since a long time (see Hall et al., 2010 for a review), our results indicate that investment in non-R&D assets also affects firm productivity. Investment in intellectual property (IP) assets (patents, copyrights, royalties, licenses), computer software, and organizational and branding capital have heterogeneous and often non-overlapping effects across firm groups. The results are reported in Table 4.

Among all types of investment in KBC, firm productivity is mostly sensitive to investment in computer software: the result in the aggregate sample is very large and more than proportional, with an elasticity of 1.6. Interestingly, this effect is confined to a few specific subsets of firms, as it appears to be driven by foreign-owned firms, firms in services, firms which export and firms that are of medium size (50 to 250 employees). Notwithstanding the different methodology and data used,³⁹ our sectoral results are comparable to those obtained by Riley and Robinson (2011) who find a larger effect of IT capital in services than in manufacturing.

In line with a large literature, we find that investment in R&D is positively associated with productivity. With the exception of the subsample of Irish firms, the related coefficient is lower compared to that of investment in software, with

³⁷This result is likely to also reflect the ability of larger firms to attract better inputs and better ³⁸For a review of this evidence see, for example, Greenaway and Kneller (2007).

³⁹Riley and Robinson (2011) measure IT capital with the number and wages of workers in IT occupations.

an elasticity ranging between 2 and 0.3. It is noteworthy that investment in R&D tends to be positively linked to productivity in those firm groups where investment in software is not significantly associated with productivity: Irish-owned and manufacturing firms. Investment in R&D is also associated with the productivity of exporters, while having no effect on non-exporters. The finding that investments in software and in R&D seem to have opposing effects on the productivity of different firms might suggest that the two intangible assets are partly substitutable with each other, and that different firms might find it optimal to invest in either of the two assets, but not both. We investigate this aspect in the next section.

Related to investment in R&D is investment in IP, as both assets are classified as innovative property assets (Corrado *et al.*, 2005). Previous studies (Crass and Peters, 2014; Higón *et al*, 2017) found no significant association between investing in IP assets (specifically patents) and firm productivity. While we confirm this finding when analyzing all firms (the coefficient is only modestly significant), when unpacking the effect across firm groups, we find that in foreign owned and manufacturing firms productivity is strongly associated with IP investment, unlike Irish and services firms, where the coefficient is close to zero. In addition, also medium-sized firms, in contrast to small firms moreover and large firms on the other, are found to benefit from investing in IP assets. Although our results might not be directly comparable to the literature because of the peculiar features of our data, they again highlight the importance of considering heterogeneous effects when analyzing the impact of intangibles on productivity.

Investment in organizational and branding capital is strongly associated with the productivity gains, with most of the firm subgroups reporting a positive and significant coefficient. The estimated elasticity of this asset ranges between 0.2 and 0.5, with the most affected firms being those with foreign ownership, active in services sectors, and having more than 250 employees. Interestingly, among the size groups, large firms are the only ones where organizational capital investment appears linked to higher productivity: this highlights that, beyond a certain scale of operations, investing in managerial skills can be a crucial factor for productivity gains. In our data, organizational practices and marketing investment are bundled together in one variable; however, some comparison with previous studies can be made. Riley and Robinson (2011) also find organizational practices⁴⁰ having a positive impact on productivity, even exceeding that of software and R&D. Crass and Peters (2014) and Higón et al. (2017) unambiguously confirm that investment in advertising has a positive impact on productivity, in all firms. In contrast, in our analysis the productivity of manufacturing firms appears to be linked to investment in organizational and branding capital in a weaker way than that of services firms.

Finally, investment in other intangible assets (not classified in one of the above-mentioned categories) is found to be associated with higher productivity particularly in firms with foreign ownership and with more than 250 employees. It is difficult to comment on this result, since we are unable to identify which assets fall in this category. What appears interesting, however, is that there might be a

 $^{^{40}\}mbox{They exploit}$ the number and wages of organizational workers, that is, managers and marketing workers.

^{© 2020} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

tendency for foreign and large firms to ascribe a relevant fraction⁴¹ of their intangibles investment to this residual group.

5.3. Complementarity/Substitution Effects of Investment in Multiple Intangible Assets

Estimates of complementarity/substitutability of pair-wise combinations of investments in intangible assets are presented in Table 5.42 Important interaction effects are found for investment in the various intangible assets, with heterogeneous effects across the subsamples of firms.

Before proceeding to the exposition of the findings, a word of caution is needed. Notice that, in Table 5, the Hansen test of over-identifying restrictions upholds the null of instrument validity; however, the *p*-values equal to one are too high for the test to be considered meaningful.⁴³ The large number of regressors which we instrument (for reverse causality concerns) in the regressions based on specification (7), determine a rapid increase of the moment conditions in the GMM estimation, which weaken the power of the test (Bowsher, 2002). Unfortunately, a large number of (highly correlated) instruments could be problematic also for the consistency of the estimates (Ziliak, 1997; Newey and Smith, 2004). In Table 5, despite collapsing the instruments as suggested by Roodman (2009), the Hansen p-values do not allow us to eliminate the concerns linked to instrument over-proliferation. For this reason, we use alternative strategies to reduce the number of instruments, based on the principal component analysis (PCA) suggested by Bontempi and Mammi (2015).44 The PCA strategy also yields Hansen *p*-values of one and does not inform our choice of which results are to be considered superior. We are aware that the identification of complementarities is a demanding exercise given the available data we analyze, and therefore these results should be interpreted as suggestive only. Even if our approach is more conservative than that proposed by Carree et al. (2011), our estimates are not completely free of concerns linked to instrument validity. Taking into account these concerns, we report the obtained complementarity/substitution estimates by imposing a further restriction: while we present the estimates obtained following the Roodman (2009) methodology (which yields results that are preferable to the PCA approach for the regressions results presented in Tables 3 and 4), we only interpret the estimated coefficients which hold⁴⁵ across both the estimation based on collapsing the instruments (reported in Table 5) and the principal component analysis (reported in Table 5B and C in the Technical Appendix).

The results suggest that investment in R&D is an important driver of the link between investment in intangibles and productivity. Firms that invest in R&D and simultaneously in other (non-R&D) intangibles have higher productivity returns from their investment in IP assets and in organizational capital. Investing in R&D

⁴⁵Have same sign and significance, these coefficients are presented in bold.

 $^{^{41}}$ For their productivity. 42 Table 5 shows only the main model coefficients and the interactions which turn out to be significant. ⁴³The Hansen test suffers from a severe under-rejection problem in case of instrument

proliferation.

These results are presented in a Technical Appendix available from the authors upon request.

appears to enhance the productivity returns of investment in IP assets in small firms and non-exporters; whereas in medium-sized firms investment in R&D is found to strengthen the impact of investment in organizational capital on productivity.⁴⁶ These results point to the crucial importance of investment in R&D for building a knowledge base that enable firms to exploit more efficiently other inputs, as shown among others by Teece (1986), Griffith *et al.* (2004), and Belderbos *et al.* (2014).

Investment in IP assets is also found to be an important complement of investment in other intangibles. Specifically, investing in IP assets increases the productivity returns of investment in R&D and in organizational and branding capital. The complementarity of investment in IP assets and in R&D is found for services firms, while in non-exporting firms, investing in IP assets is complementary to investment in organizational capital. These findings highlight the importance of prior knowledge, embedded in patents, licenses and copyrights, in enhancing the absorptive capacity of firms (Cohen and Levinthal, 1990) which make investment in R&D, management and marketing more productive (Crass and Peters, 2014). Furthermore, in the case of firms in services, productivity returns to investment in R&D are enhanced by investment in software. This result is consistent with recent evidence on complementarities between investments in R&D and information and communication technologies provided by Corrado *et al.* (2017) and Mohnen *et al.* (2018).

In addition to uncovering the complementarity of investments in R&D and IP assets, this paper makes a further contribution to the literature suggesting that investments in organizational and branding capital might substitute investments in other intangibles. It is worth noticing that investment in organizational capital appears to be substitutable to investments in intangibles which, for specific firm groups, are not significantly associated with firm productivity (for a comparison see Table 4). Investment in organizational capital is a substitute for investment in R&D in the case of exporters and in medium sized firms, and a substitute for IP assets in non-exporting firms.⁴⁷ We consider this finding interesting, because it suggests that while investment in organizational and branding capital has an overall positive effect on productivity, it could be a substitute for investment in intangibles, such as R&D in the case of exporters and IP assets in the case on non-exporters. which appear to have no significant effect on productivity in these firm groups. To the best of our knowledge, this evidence is new to the literature. Further research could shed light on the robustness of this result in the context of other countries' economic structures and institutional frameworks.

⁴⁶In one case, for non-exporters, R&D seems to decrease the effectiveness of another intangible, namely organizational capital.

⁴⁷This peculiar result of organizational capital being a substitute of investments which, themselves, are not found to be associated with productivity gains, applies also for a number of other cases (for R&D investment of foreign, service and small firms; for IP investment of non-exporters). These coefficients, however, are not robust to applying the principal component analysis suggested by Bontempi and Mammi (2015).

^{© 2020} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

		Inve	ESTMENT IN A	T, ll Intangib	ABLE 3 LE Assets and Firm	PRODUCTIVITY				
	(1)	(2)	(3)	(4)	(5)	(9)	(7)	(8)	(6)	(10)
Dependent Variable					Ln (VA/Em	(ployee)				
Sample	All Firms	Irish	Foreign	Services	Manufacturing	Non- Exporters	Exporters	Small	Medium	Large
Ln(VA/Employee) _{t-1}	0.544***	0.510***	0.561***	0.561***	0.502***	0.469***	0.620***	0.520***	0.546***	0.566***
Ln(Intangibles/Employee)	(0.276**	(0.040) 0.287	(0.009) 0.374^{***}	(c/0.0) 0.293*	(0.022) 0.294***	(90.0)	(0.000)	(0.196^{**})	(0.193*)	(0.104) 0.177
I n(Tannihles/Emnloyee)	(0.140) -0.078	(0.254)	(0.113)	(0.174)	(0.069) 0.007	(0.183)	(0.132)	(0.098) 0.079	(0.116)	(0.281)
Lu(I augures Lupto) ((0.124)	(0.168)	(0.151)	(0.155)	(0.139)	(0.171)	(0.152)	(0.198)	(0.145)	(0.180)
Ln(Wage/Employee)	-0.251	-0.115	-0.487	-0.117	0.102	0.548	-0.916^{\dagger}	0.555	-0.139	0.051
L'n(Aœ)	(0.234) 0.020**	(0.484) 0 019***	(0.474) 0.031	(0.290) 0.073**	(0.275) 0.009	(0.345) 0.076***	(0.612) 0.009	(0.340) 0 017 [†]	(0.412) 0 075**	(0.888) 0.015
	(0.000)	(0.007)	(0.025)	(0.010)	(0.010)	(0.00)	(0.016)	(0.008)	(0.012)	(0.037)
Foreign-owned	0.098^{***}	~	~	0.072^{**}	0.104^{***}	0.062^{**}	0.117^{***}	0.103^{***}	0.072^{**}	0.074
I n(Market Share)	(0.022)	1 375†	0355	(0.029)	(0.021) 1 154**	(0.030) 1 152 [†]	(0.024) 0.946	(0.030)	(0.031) 0.880	(0.054)
	(0.623)	(0.918)	(0.839)	(166.0)	(0.549)	(0.774)	(0.658)	(2.619)	(1.069)	(0.806)
Exporter	0.040***	0.046***	0.007	0.037**	0.025***			0.019	0.030	0.080*
Constant	(110.0) -0.001	(0.014) -0.017 [†]	(0.041) 0.107**	(0.016) 0.014	(/00.0) 	-0.040^{***}	0.066^{**}	(0.012) - 0.009	(0.024) 0.007	(0.042) - 0.017
	(0.011)	(0.011)	(0.050)	(0.014)	(0.010)	(0.012)	(0.032)	(0.010)	(0.023)	(0.042)
Time FE	Υ	Υ	Υ	Υ	Υ	Υ	Y	Υ	Υ	Υ
NACE 2-dig FE Firm FE	YY	YY	YY	YY	Y	YY	YY	YY	YY	YY
N observations	25674	20729	4945	17336	7809	16405	9269	14560	6906	2045
GMM instr.	39	38	38	39 	39 . 2 <u>9</u>	38	38 38	39	39 26-	39
<i>P</i> value AK2 test <i>P</i> value Hansen test	$0.964 \\ 0.200$	0.734 0.366	0.980 0.228	0.635 0.722	0.379 0.993	0.469 0.275	0.298 0.568	0.468 0.273	$0.767 \\ 0.214$	$0.692 \\ 0.168$
Notes: Estimates are obta	ined with a s	system-GMN	A estimator.	We instrum	ent all the continuor	us variables, ex	cept age, with	n lagged leve	als dated from	n t-2 to t-6 in

the equation in difference and lagged differences dated *t*-1 in the equation in levels. Instruments are collapsed following Roodman (2009), in order to prevent estimates distortion arising from instrument over-proliferation (see Technical Appendix for details). Standard errors clustered at the NACE 2-digit sector level in parentheses p < 0.15, p < 0.10, **p < 0.01, **p < 0.01. *Source:* Authors' calculations based on data from on CIP and ASI data.

Review of Income and Wealth, Series 67, Number 2, June 2021

	PRODUCTIVITY
	Firm
	AND]
4	ASSETS .
[ABLE	GIBLES .
	INTAN
	ECIFIC
	IN SP
	MENT I
	INVEST

	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)
Dependent Variable					Ln (VA/E	(mployee)				
Sample	All Firms	Irish	Foreign	Services	Manufacturing	Non-Exporters	Exporters	Small	Medium	Large
Ln(VA/Employee) _{t-1}	0.513***	0.485***	0.515***	0.494***	0.485*** (0.050)	0.475*** (0.049)	0.587***	0.520***	0.609***	0.475***
Ln(Software/	1.610^{***}	2.213	1.648^{***}	2.246*	0.761	1.000	1.607^{***}	-1.379^{\ddagger}	3.427*	0.937
Employee)	(195.0)	(1 842)	(0 574)	(1176)	(0,616)	(1 924)	(0 541)	(0 924)	(0.030)	(0.717)
Ln(R&D/Employee)	0.102	2.149***	0.107	-0.164	0.286**	-0.071	0.354*	0.016	0.173	-0.399
Ln(IP Assets/	(0.255) 0.130^{\dagger}	(0.807) 0.092	(0.226) 0.171^{**}	(0.735) 0.008	(0.139) 0.263^{***}	(0.308) 0.334	(0.185) 0.071	(0.321) 0.046	(0.218) 0.243***	(0.407) -0.070
Employee)						i				
I n(Ora & Brandina/	(0.088) 0 350***	(0.254)	(0.077)	(0.129)	(0.036)	(0.285) 0.108**	(0.105)	(0.198)	(0.055)	(0.141) 0 576***
Empl)	6000	+00.0	0.4.0	700.0	0.174	06110		0.121	761.0	070.0
	(0.092)	(0.246)	(0.117)	(0.109)	(0.122)	(660.0)	(0.132)	(0.218)	(0.123)	(0.189)
Ln(Other Intang./ Employee)	1.129**	0.001	1.356***	1.275**	0.599*	-0.326	0.304	0.020	-0.104	0.530^{**}
	(0.518)	(0.252)	(0.474)	(0.558)	(0.357)	(0.727)	(0.265)	(0.329)	(0.207)	(0.266)
Ln(Tangibles/ Employee)	<u>-0.126</u>	-0.07Ź	-0.260^{*}	-0.104	-0.023	-0.090	-0.188 [†]	0.092	-0.204*	<u>-0.056</u>
	(0.153)	(0.151)	(0.153)	(0.158)	(0.103)	(0.156)	(0.125)	(0.149)	(0.115)	(0.178)
Ln(Wage/Employee)	0.209	0.112	0.130	0.566	0.190	0.735	-0.021	0.923°	-0.059	1.291
Ln(Age)	0.014^{*}	0.020**	0.018	0.014	(+1.5.0)	0.021***	-0.003	(200.0) 0.008	(0.023)	-0.010
Eomion annad	(0.007)	(0.008)	(0.025)	(0.011)	(0.012)	(0.007)	(0.015)	(0.007)	(0.012)	(0.035)
rotetgii-owited	(0.019)			(0.031)	(0.019)	0.045 (0.036)	(0.020)	(0.028)	(0.027)	(0.041)
Ln(Market Share)	0.851	0.276	0.535	1.309	1.345**	1.938*	0.369	3.955	0.787	-0.211
Evnortor	(0:050) 0.020***	(0.842)	(/9/.0)	(1.258)	(0.669)	(/.c0.1)	(0./08)	(2.634)	(10/.0)	(0.727)
EAPUTG	(0.011)	(0.012)	(0.033)	(0.017)	(0.008)			(0.017)	(0.018)	(0.039)

				TABL	E4 (CONTINUE	D)				
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)
Dependent Variable					Ln (VA/F	Employee)				
Sample	All Firms	Irish	Foreign	Services	Manufacturing	Non-Exporters	Exporters	Small	Medium	Large
Constant	-0.004 (0.010)	-0.007 (0.014)	0.129 *** (0.032)	-0.009 (0.016)	-0.021^{**} (0.010)	-0.045^{**} (0.018)	0.034^{*} (0.021)	-0.018 (0.016)	-0.018 (0.013)	-0.063^{\dagger} (0.039)
Time FE NACE 2-dig FE Firm FE	***	XXX	$\mathbf{X}\mathbf{X}$	$\mathbf{X}\mathbf{X}\mathbf{X}$	XXX	XXX	ХХХ	XXX	ХХХ	XXX
N observations No. GMM instr.	25674 63	20729 62	4945 62	$17336 \\ 63$	7809 63	16405 62	9269 62	$14560 \\ 63$	9069 63	2045 63
<i>P</i> value AR2 test <i>P</i> value Hansen test	0.657 0.303	$0.574 \\ 0.323$	$0.516 \\ 0.412$	$0.486 \\ 1.000$	0.449 1.000	0.426 0.501	$\begin{array}{c} 0.501 \\ 0.386 \end{array}$	$0.369 \\ 0.315$	$0.775 \\ 0.236$	$0.952 \\ 0.637$
<i>Notes:</i> Estimates at the equation in differenc distortion arising from i $\int_{0}^{\infty} c (0.15^{+})^{*} c = 0.10$	e obtained wi e and lagged d nstrument ove	th a system- lifferences d r-proliferati	GMM estimated t-1 in the on (see Techr	ator. We insi e equation ir nical Appenc	rument all the cont levels. Instruments lix for details). Stan	inuous variables, ex are collapsed follov dard errors clustere	cept age, with ving Roodman d at the NACE	lagged level (2009), in c 2-digit sect	s dated from order to preve	<i>t</i> -2 to <i>t</i> -6 in at estimates centheses.

p < 0.15, p < 0.10, p < 0.05, p < 0.01. Source: Authors' calculations based on data from on CIP and ASI data.

	INVESTMEN	ut in Intanc	SIBLES AND P	RODUCTIVITY	: INTERDEPENDENC	e Patterns by Firm	1 SIZE AND GF	SOUPS		
	(1)	(2)	(3)	(4)	(5)	(9)	(7)	(8)	(6)	(10)
Dependent Variable					Ln (VA/	Employee)				
Sample	All firms	Irish	Foreign	Services	Manufacturing	Non-Exporters	Exporters	Small	Medium	Large
Ln (VA/Empl),-1	0.506***	0.531***	0.516***	0.529***	0.463***	0.427***	0.556***	0.520***	0.611***	0.523***
Ln (Soft/Emp)	(0.045) 2.453 [†]	(0.00) -3.531	(0.0/7) 1.876*	(0.066) 2.109	(0.02) 6.254	(0.00) 0.992	(0.00) (0.015)	(9.0.0) -9.550	(0.00/) 5.904***	(0.072) 3.050
	(1.615)	(3.772)	(1.047)	(3.061)	(12.525)	(1.316)	(1.926)	(6.940)	(1.032)	(2.294)
Ln (R&D/Emp)	0.377	1.123^{**}	0.270	0.184	0.501^{***}	-0.282	0.380	0.324	0.201	-0.449
	(0.311)	(0.552)	(0.283)	(0.398)	(0.177)	(1.011)	(0.325)	(0.871)	(0.190)	(0.752)
Ln (IP Ass./Emp)	0.1231	0.431	0.165^{**}	0.047	0.274***	0.039	0.099	-0.040	0.220^{**}	0.027
In (Ora Can/Emn)	(0.081) 0 285 $***$	(0.263)	(0.082) 0 787***	(0.047) 0 130*	(0.059) 0.184	(0.068) 0 307***	(0.121)	(0.061) 0 215***	(0.095) (0.065)	(0.131) 0 311**
Lu (Org Capi Lunp)	(0.065)	(0.267)	(0.073)	(0.072)	(0.593)	(0.112)	(0.108)	(0.074)	(0.132)	(0.158)
Ln(Soft/Emp) * R&D	0.194	2.364	-0.176	-2.276	-0.526	-1.070	-1.605	-6.484	3.547	-0.129
× 4	(1.024)	(1.980)	(1.311)	(2.326)	(1.936)	(2.782)	(2.194)	(6.808)	(2.225)	(0.780)
Ln(Soft/Emp) * Org Cap	-0.692	1.893	-0.371	1.082	1.096	-1.981	1.843	7.697	-6.449**	-2.008
	(1.409)	(2.685)	(1.366)	(1.812)	(2.588)	(2.401)	(2.451)	(7.593)	(2.521)	(1.437)
Ln(Sott/Emp) * IP Ass.	(20.02)	1.461	-0.082	-0.054 (7 887)	COC.C-	3. /00° (7 767)	0.009	(LLC L)	-2.414	0.237
Ln(R&D/Emp) * Soft.	-0.095	-0.332	-0.135	0.604***	-0.233	-0.086	-0.175	-0.531^{\dagger}	-0.136	0.530
	(0.255)	(0.646)	(0.223)	(0.184)	(0.385)	(1.111)	(0.326)	(0.327)	(0.310)	(0.419)
Ln(R&D/Emp) * Org Cap	-0.264^{**}	-0.346	-0.305^{**}	-0.762^{***}	-0.235*	0.508	-0.191^{**}	-0.410^{*}	-0.381***	0.496
	(0.115)	(0.743)	(0.126)	(0.290)	(0.122)	(0.950)	(0.082)	(0.213)	(0.124)	(0.879)
Ln(K&U/Emp) * IF ASS.	/ 50.0	0.040	0.131	0.0/4	071.0	0.145	07070	0.00/	0.214	100.0-
Ln(Org Cap/Emp) * Soft	-0.056	(0.00)	(0.230) 0.035	(c81.0) 0.486 [†]	(0.100) -0.199	(0.833) 0.155	(0.298) -0.048	(0.916) 0.189	(0.230) 0.132	(1.083) - 0.191

TABLE 5 Ě

				TABLE 5	CONTINUED	()				
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)
Dependent Variable					Ln (VA/	Employee)				
Sample	All firms	Irish	Foreign	Services	Manufacturing	Non-Exporters	Exporters	Small	Medium	Large
Ln(Org Cap/Emp) * R&D	(0.219) -0.153 (0.161)	(0.959) -0.139 (0.423)	(0.202) -0.060 (0.114) (0.114)	$\begin{pmatrix} (0.299) \\ 0.046 \\ (0.178) \\ 0.017 \\ 0.178 \end{pmatrix}$	(0.333) 0.007 (0.090)	(0.407) -0.513** (0.240)	(0.153) (0.082) (0.104)	$\begin{pmatrix} 0.262 \\ -0.174 \\ (0.259 \end{pmatrix}$	(0.202) 0.303** (0.154)	(0.220) -0.024 (0.201)
Ln(Urg Cap/Emp) * 1P Ass.	0.411*	-0.632	0.312	0.216	0.185	0.496*	0.024	-0.064	-0.160	0.233
Ln(IP Ass/Emp) * Soft	(0.243) -0.056	(0.528) -0.162	(0.202) 0.014 (0.162)	(0.293) 0.021 (0.074)	(0.751) 0.208^{***} (0.067)	(0.292) -0.097 (0.162)	(0.247) 0.025 (0.121)	(0.245) 0.116 (0.253)	(0.147) 0.396^{***}	(0.388) -0.062
Ln(IP Ass./Emp) * R&D	0.052	-0.230	-0.010	0.076	0.005	0.386**	-0.089^{\dagger}	0.420**	-0.027	-0.068
Ln(IP Ass./Emp) * Org Cap	-0.145	-0.322	-0.151	-0.245	-0.012	-0.285**	0.005	0.201**	0.023	-0.021
T	(0.130)	(0.369)	(0.116)	(0.175)	(060.0)	(0.141)	(0.067)	(0.101)	(0.063)	(0.085)
Other Firm Controls Time FE NACE 2-dig FE Firm FE	XXXX	XXXX	XXXX	XXXX	XXXX	XXXX	XXXX	XXXX	XXXX	XXXX
N observations GMM instruments <i>P</i> value AR2 test <i>P</i> value Hansen test	$25674 \\ 243 \\ 0.758 \\ 1$	$20729 \\ 242 \\ 0.386 \\ 1$	4945 242 0.776 1	$17336 \\ 243 \\ 0.728 \\ 1$	7809 243 0.955 1	16405 242 0.322 1	9269 242 0.405 1	14560 243 0.277 1	9069 243 0.408 1	$\begin{array}{c} 2045 \\ 243 \\ 0.918 \\ 1 \end{array}$
<i>Notes:</i> Estimates are obta the equation in difference and distortion arising from instrur Bold indicates coefficients ${}^{*}p < 0.15, {}^{*}p < 0.10, {}^{**}p < Source: Authors' calculati$	tined with a lagged different over-program $s_{s}^{***} = 0.05, s_{s}^{***} > 0.05$ ons based or	system-GN rences dated oliferation (0.01. n data from	AM estimato d t-l in the ex (see Technici t on CIP and	ur. We instru quation in le al Appendix ASI data.	ment all the contir evels. Instruments a for details). Stand:	nuous variables, exc tre collapsed follow ard errors clusterec	cept age, with ing Roodman I at the NACE	lagged level (2009), in o 3 2-digit sect	s dated from rder to prever or level in par	<i>t</i> -2 to <i>t</i> -6 in at estimates entheses.

Review of Income and Wealth, Series 67, Number 2, June 2021

@ 2020 The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

6. CONCLUSIONS AND POLICY IMPLICATIONS

This paper extends a Cobb-Douglas production function theoretical framework and provides novel evidence on productivity returns of investment in intangible assets in a small open economy. To the best of our knowledge, we make three methodological contributions to the literature. First, we model the relationship between investment in KBC and productivity as a dynamic process accounting for productivity's persistence and path-dependency. Second, we contextualize the relationship between investment in KBC and productivity to the features of a small open economy accounting for the role of exporting and inward foreign investment. Third, and related to a small open economy framework, we relax the assumption of a homogeneous behavior of a representative firm and allow the link between productivity and investment in KBC to differ across groups of firms and sectors.

The results of this analysis indicate that investment in KBC is positively associated with firm productivity. On average, *ceteris paribus*, a ten percent increase in the investment in KBC per employee is associated with a three per cent productivity gain. However, this aggregate result hides heterogeneous effects across firm groups, which in a small open economy like Ireland are more evident than in large economies.

Our estimates identify a larger responsiveness of productivity to investment in KBC for foreign-owned firms in comparison to Irish-owned firms. Across size groups, the sensitivity of productivity to investment in KBC is found to be greater for small and medium-sized firms (with less than 250 employees). A noteworthy finding of this research is that investment in KBC appears to be more important for productivity than investment in tangible assets which, over and above other factors, does not have a significant effect. A possible explanation for this observation might be the small variation over the analyzed period in investment in tangible capital, as well as the limited impact on productivity due to pre-existing large stocks of physical capital.

Investment in R&D is positively linked to firm productivity in some of the sub-samples of firms analyzed, namely indigenous firms, manufacturing firms and exporters, with no distinction found among firms with different size. The productivity of foreign-owned firms, and of those active in services sectors is mainly associated with investment in non-R&D intangible assets, including software and organizational and branding capital.

In the case of investment in non-R&D assets (IP assets, software, organizational and branding capital, and other intangibles), investment in software is most strongly associated with higher productivity (both economically and statistically). Higher investment in organizational and branding capital also appears to positively affect the productivity of the various types of firms analyzed, with distinctions in terms of firms' ownership or sector of activity being more nuanced. In contrast, investing in IP assets only affects the productivity of foreign-owned firms, medium sized firms and manufacturers.

Furthermore, this analysis finds that investments in multiple intangible assets could be complementary as well as substitutable depending on the specific combination of investments and firm and sector characteristics. One key result is that investing in R&D and IP assets enhances the productivity returns of simultaneous

^{© 2020} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

investments in other intangible assets. This result highlights the importance of investment in R&D and prior knowledge for building firms' absorptive capacity as suggested by, among others, Griffith *et al.* (2004) and Belderbos *et al.* (2014). Another novel result of this analysis is that investments in management and marketing are substitutable with investments in other intangible assets, particularly for firms where these (other) investments do not appear to matter for productivity.

Taken together, our research results indicate that productivity responds differently to investments in various types of KBC across Irish-owned and foreign-owned firms. For Irish-owned firms, productivity is most strongly linked to investment in R&D intangible assets and in organizational and branding capital. For foreign-owned firms, productivity is most strongly linked to investment in non-R&D intangible assets such as computer software, intellectual property assets and organizational and branding capital.

It is widely acknowledged that given market and systemic failures specific to knowledge-based capital,⁴⁸ firms tend to underinvest in such assets, below the socially desirable level. The empirical evidence provided in this paper suggests a number of implications for the design of policies and strategies aimed at incentivizing investment in KBC in the enterprise sector.

First, the results of this analysis indicate that a more comprehensive policy approach to incentivize investment in a broader range of intangible assets beyond R&D such as IP assets, computer software, and organizational and branding capital could be beneficial. Second, the results also indicate that incentivizing firms to invest in R&D could enhance the productivity returns of investments in other intangible assets. In contrast, over-investing in management and marketing could be counterproductive. Third, the evidence provided in this paper suggests that policy measures should be targeted to specific groups of firms with similar characteristics, such as: domestic versus foreign-owned; SMEs versus large; manufacturing versus services firms.

While this paper highlights the importance of investments in a broad range of intangible assets for firm productivity, further research could provide useful evidence. Possible research directions that could be explored include identifying factors driving firms' choice of investments in various intangible assets and how these choices affect firm performance outcomes such as exporting and employment growth.

References

Andrews, D. and A. de Serres, "Intangible Assets, Resource Allocation and Growth. A Framework for Analysis," OECD Economics Department Working Paper No. 989, 2012.

Arrighetti, A., F. Landini, and A. Lasagni, "Intangible Assets and Firm Heterogeneity: Evidence From Italy," *Research Policy*, 43(1), 202–13, 2014.

Bartelsman, E. J. and M. Doms, "Understanding Productivity: Lessons from Longitudinal Microdata," Journal of Economic Literature, 38(3), 569–94, 2000.

⁴⁸Market failures include knowledge-spillovers and information asymmetries. Systemic failures are related to co-ordination and networking costs. A recent review is provided by Andrews and de Serres (2012).

- Battisti, M., F. Belloc, and M. Del Gatto, "Unbundling Technology Adoption and *tfp* at the Firm Level: Do Intangibles Matter?," *Journal of Economics & Management Strategy*, 24, 390–414, 2015.
- Baum, C. F., H. Lööf, Laijani P. Nabavi, and A. Stephan, "A new approach to estimation of the R&Dinnovation-productivity relationship," *Economics of Innovation and New Technology*, 26(1–2), 121–33, 2017.
- Belderbos, R., B. Lokshin, and B. Sadowski, "The returns to foreign R&D," Journal of International Business Studies, 46(4), 491–504, 2014.
- Bontempi, M. and J. Mairesse, "Intangible capital and productivity at the firm level: A panel data assessment," *Economics of Innovation and New Technology*, 24(1–2), 22–51, 2015.
- Bontempi, M., and I. Mammi, "Implementing a strategy to reduce the instrument count in panel GMM," *Stata Journal*, 15(4), 1075–97, 2015.
- Bowsher, C. G., "On Testing Overidentifying Restrictions in Dynamic Panel Data Models," *Economics Letters*, 77, 211–20, 2002.
- Blundell, R. and S. Bond, "Initial Conditions and Moment Restrictions in Dynamic Panel Data Models," *Journal of Econometrics*, 87(1), 115–43, 1998.
- Bond, S. (2002). "Dynamic Panel Data Models: a Guide to Microdata Methods and Practices," CEMMAP Woking Paper CWP09/02.
- Broström, A. and S. Karlsson, "Mapping Research on R&D, Innovation and Productivity: A Study of an Academic Endeavour," *Economics of Innovation and New Technology*, 26(1–2), 6–20, 2017.
- Carree, M., B. Lokshin, and R. Belderbos, "A Note on Testing for Complementarity and Substitutability in the Case of Multiple Practices," *Journal of Productivity Analysis*, 35(3), 263–269, 2011.
- Chappell, N. and A. Jaffe, "Intangible investment and firm performance," *Review of Industrial Organization*, 52(4), 509–55, 2018.
- Chen, W., "Cross-Country Income Differences Revisited: Accounting for the Role of Intangible Capital," *Review of Income and Wealth*, 64(3), 626–48, 2018.
- Cohen, W. M., and D. A. Levinthal, "Absorptive Capacity: A New Perspective on Learning and Innovation," Administrative Science Quarterly, 35(1), 128–152, 1990.
- Corrado, C., C. Hulten, and D. Sichel, "Measuring Capital and Technology: An Expanded Framework," in C. Corardo, J. Haltiwanger, and D. Sichel (eds), *Measuring Capital in a New Economy*, NBER and University of Chicago Press, Chicago IL, 11–46, 2005.
- _____, "Intangible Capital and US Economic Growth," *Review of Income and Wealth*, 55(3), 661–85, 2009.
- Corrado, C., J. Haskel, and C. Jona-Lasinio, "Knowledge Spillovers, ICT and Productivity Growth," Oxford Bulletin of Economics and Statistics, 79(4), 592–618, 2017. https://doi.org/10.1111/ obes.12171.
- Corrado, C., J. Haskel, C. Jona-Lasinio, and M. Iommi, Intangible Capital and Growth in Advanced Economies: Measurement Methods and Comparative Results", INTAN-Invest. 2012. www. INTAN-Invest.net.
 - _____, "Intangibles and Industry Productivity Growth: Evidence from the EU", INTAN-Invest, 2014. www.INTAN-Invest.net.
- , "Intangible Investment in the EU and US Before and Since the Great Recession and its Contribution to Productivity Growth," *Journal of Infrastructure, Policy and Development*, 2(1), 11–36, 2018.
- Crass, D. and B. Peters, "Intangible Assets and Firm-Level Productivity," Centre for European Economic Research, Mannheim, Discussion Paper No. 14-120, 2014.
- Crépon, B., E. Duguet, and J. Mairesse, "Research, Innovation, and Productivity: An Econometric Analysis at the Firm Level," *Economics of Innovation and New Technology*, 7, 115–56, 1998.
- Dal Borgo, M., P. Goodridge, J. Haskel, and A. Pesole, "Productivity and Growth in UK Industries: An Intangible Investment Approach," Oxford Bulletin of Economics and Statistics, 75(6), 806–34, 2013.
- Eaton, J., and S. Kortum, "Technology, Geography, and Trade," Econometrica, 70(5), 1741-79, 2002.
- Foster, L., J. Haltiwanger, and C. Syverson, "Reallocation, Firm Turnover, and Efficiency: Selection on Productivity or Profitability?," *American Economic Review*, 98(1), 394–425, 2008.
 Greenaway, D. and R. Kneller, "Firm Heterogeneity, Exporting and Foreign Direct Investment,"
- Greenaway, D. and R. Kneller, "Firm Heterogeneity, Exporting and Foreign Direct Investment," *Economic Journal*, 111, F134–F161, 2007.
- Griffith, R., S. Redding, and J. Van Reenen, "Mapping the Two Faces of R&D: Productivity Growth in a Panel of OECD Industries," *The Review of Economics and Statistics*, 86(4), 883–95, 2004.
- Griliches, Z., "Issues in Assessing the Contribution of Research and Development to Productivity Growth," *Bell Journal of Economics*, 10(1), 92–116, 1979.

, "R&D and the Productivity Slowdown," *American Economic Review, Proceeding Issue*, 70(2), 343–8, 1980.

^{© 2020} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth

, "R&D and Productivity at Firm Level," in Z. Griliches (ed.), *R&D and Productivity: The Econometric Evidence*, National Bureau of Economic Research. University of Chicago Press, Chicago, IL, 100–33, 1998.

- Hall, B. H., "Innovation and Productivity," Nordic Economic Policy Review, 2, 167-204, 2011.
- Hall, B. H., F. Lotti, and J. Mairesse, "Evidence on the Impact of R&D and ICT Investments on Innovation and Productivity in Italian Firms," *Economics of Innovation and New Technology*, 22(3), 300–28, 2013.
- Hall, B. H., J. Mairesse, and P. Mohnen, "Measuring the Returns to R&D." in Hall, B. H., and N. Rosenberg, (eds), *Handbook of the Economics of Innovation*. vol. 2. Elsevier, Amsterdam, 1033– 1082, 2010.
- Higón, A. D., J. Gómez, and P. Vargas, "Complementarities in Innovation Strategy: Do Intangibles Play a Role in Enhancing Firm Performance?," *Industrial and Corporate Change*, 26(5), 865–86, 2017. https://doi.org/10.1093/icc/dtw055.
- Hopenhayn, H. A., "Entry, Exit, and Firm Dynamics in Long Run Equilibrium," *Econometrica*, 1127– 50, 1992.
- Hulten, C., "On the Importance of Productivity Change," American Economic Review, 69, 126–36, 1979.
- Jona-Lasinio, C. and V. Meliciani, "Global Value Chains and Productivity Growth in Advanced Economies: Does Intangible Capital Matter?," *International Productivity Monitor*, 36, 53–78, 2019.
- Karlsson, C., G. Maier, M. Trippl, I. Siedschlag, R. Owen, and G. Murphy, "ICT and Regional Economic Dynamics: A Literature Review," *European Commission, Joint Research Centre, Institute* for Prospective Technological Studies, Publications Office of the European Union, Luxembourg, 2010.
- Lucas, R. E., "On the Mechanics of Economic Development," *Journal of Monetary Economics*, 22(1), 3–42, 1988.
- Mairesse, J., and P. Mohnen, "Using Innovation Surveys for Econometric Analysis." in Hall, B. H., and N. Rosenberg, (eds), *Handbook of the Economics of Innovation*. vol. 2. Elsevier, Amsterdam, 1130–55, 2010.
- Mankiw, G., D. Romer, and D. Weil, "A Contribution to the Empirics of Economic Growth," *Quarterly Journal of Economics*, 107, 407–37, 1992.
- Marrocu, E., R. Paci, and M. Pontis, "Intangible Capital and Firms' Productivity," *Industrial and Corporate Change*, 21(2), 377–402, 2012.
- Melitz, M. J., "The Impact of Trade on Intra-Industry Reallocations and Aggregate Industry Productivity," *Econometrica*, 71, 1695–725, 2003.
- Mohnen, P., M. Polder, and G. van Leeuwen, "ICT, R&D and Organisational Innovation: Exploring Complementarities in Investment and Production," NBER Working Paper No. 25044, 2018.
- Moreira, S., "Firm Dynamics, Persistent Effects of Entry Conditions, and Business Cycles," Working Papers 17–29, Center for Economic Studies, U.S. Census Bureau, Washington DC, 2017.
- Newey, W. K., and R. J. Smith, "Higher-Order Properties of GMM and Generalized Empirical Likelihood Estimators," *Econometrica*, 72, 219–55, 2004.
- Nickell, S., "Biases in Dynamic Models with Fixed Effects," Econometrica, 49(6), 1417-26, 1981.
- Niebel, T., M. O'Mahony, and M. Saam, "The Contribution of Intangible Assets to Sectoral Productivity Growth in the EU," *Review of Income and Wealth*, 63(1), S49–S67, 2017.
- OECD, Supporting Investment in Knowledge Capital, Growth and Innovation, OECD Publishing, Paris, 2013. https://doi.org/10.1787/9789264193307-en.
- Pakes, A., and Z. Griliches, "Patents and R&D at the Firm Level: A First Look." in Griliches, Z., (ed), R&D, Patents, and Productivity. University of Chicago Press, Chicago, 55–72, 1984.
- Piva, M. and M. Vivarelli, "Is Demand-Pulled Innovation Equally Important in Different Groups of Firms?," *Cambridge Journal of Economics*, 31, 691–710, 2007.
- Riley, R. and C. Robinson, "Skills and Economic Performance: The Impact of Intangible Assets on UK Productivity Growth," UK Commission for Employment and Skills, Evidence Report No 39, 2011.
- Roth, F., "Intangible Capital and Labour Productivity Growth: A Review of the Literature," Hamburg Discussion Papers in International Economics (No. 4), 2019.
- Roodman, D., "A Note on the Theme of Too Many Instruments," Oxford Bulletin of Economics and Statistics, 71, 135–58, 2009.
- Romer, P. M., "Human Capital and Growth: Theory and Evidence," Carnegie-Rochester Conference Series on Public Policy, 32, 251–86, 1990.
- Ruane, F. P. and I. Siedschlag, "Boosting Innovation and Productivity in Enterprises: What Matters? What Works?," in Pete Lunn and Frances Ruane (eds), Using Evidence to Inform Policy, Gill & MacMillan, Dublin, 96–117, 2013.

Siedschlag, I. and X. Zhang, "Internationalisation of Firms and Their Innovation and Productivity," *Economics of Innovation and New Technology*, 24(3), 183–203, 2015.

Syverson, C., "Market Structure and Productivity: A Concrete Example," *Journal of Political Economy*, 112(6), 1181–222, 2004a.

, "Product Substitutability and Productivity Dispersion," *Review of Economics and Statistics*, 86(2), 534–50, 2004b.

____, "What Determines Productivity?," Journal of Economic Literature, 49(2), 326–65, 2011.

Teece, D. J., "Profiting from Technological Innovation: Implications for Integration," *Collaboration, Licensing and Public Policy, Research Policy*, 15(6), 285–305, 1986.

Ziliak, J. P., "Efficient Estimation with Panel Data When Instruments are Predetermined: An Empirical Comparison of Moment Condition Estimators," *Journal of Business and Economic Statistics*, 15, 419–31, 1997.

Weitzman, M. L., "On the Welfare Significance of National Product in a Dynamic Economy," *Quarterly Journal of Economics*, 90, 156–62, 1976.

SUPPORTING INFORMATION

Additional supporting information may be found in the online version of this article at the publisher's web site:

Table A1. Description of variables**Table A2.** Descriptive statistics for regression variables

^{© 2020} The Authors. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth